Author’s response to reviews

Title: Safety of a bivalent, killed, whole-cell oral cholera vaccine in pregnant women in Bangladesh: Evidence from a randomized placebo-controlled trial

Authors:

Ashraful Khan (ashrafk@icddrb.org)
Md Ali (mali25@jhu.edu)
Alamgir Kabir (alamgir.kabir@icddrb.org)
Arifuzzaman Khan (arifkhan@icddrb.org)
Md Taufiquil Islam (taufiquilislam@icddrb.org)
Afroza Akter (afrozaakter@icddrb.org)
Fahima Chowdhury (fchowdhury@icddrb.org)
Amit Saha (amit@icddrb.org)
Iqbal Khan (iqbalansary@gmail.com)
Nirod Saha (nirod@icddrb.org)
John Clemens (jclemens@icddrb.org)
Julia Lynch (Julia.Lynch@ivi.int)
Jean Excler (JeanLouis.Excler@ivi.int)
Sachin Desai (sach_des@yahoo.com)
Deok Kim (drkim@ivi.int)
Ajit Singh (ajitsingh146@hotmail.com)
Deok Ryun Kim (drkim@ivi.int)
Firdausi Qadri (fqadri@icddrb.org)

Version: 4 Date: 01 Apr 2019

Author’s response to reviews:
1. On the title page please indicate the who is the corresponding author and their relevant contact information.

Response: Details of corresponding author has been indicated.

2. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

Response: Contact email has been mentioned.

3. The Author contribution section includes reference to MAK as an author, who is not listed as an author elsewhere. Please clarify whether MAK should be included as an author for this manuscript.

If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship

Response: MAK is the abbreviation of Md. Arifuzzaman Khan. Author list has been corrected accordingly.

4. We note that two authors have the same initials (AK). To distinguish between them in the Authors’ Contributions section please designate them as AK1 and AK2, with AK1 corresponding to Alamgir Kabir, chosen because this name is furthest up the author list.

Response: Response to question-3 will remove the confusion. Now there is one AK as initial in the manuscript.

5. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.

Response: Now removed.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

BMC Infectious Diseases operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Response: As you requested to use any reputable English language editing service for language correction, we have edited the manuscript by American Journal Experts (http://bit.ly/AJE_BS). We have done and submitted the revised article during last resubmission. Now we have attached certificate from the editing service.

---------------------

Editorial Policies

---------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Infectious Diseases must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal’s Submission Guidelines web
Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

Further information about our editorial policies can be found at the following links:

Ethical approval and consent:

http://www.biomedcentral.com/about/editorialpolicies#Ethics

Availability of data and materials section:

http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials
Recipients of this email are registered users within the Editorial Manager database for this journal. We will keep your information on file to use in the process of submitting, evaluating and publishing a manuscript. For more information on how we use your personal details please see our privacy policy at https://www.springernature.com/production-privacy-policy. If you no longer wish to receive messages from this journal or you have questions regarding database management, please contact the Publication Office at the link below.

In compliance with data protection regulations, you may request that we remove your personal registration details at any time. (Use the following URL: https://www.editorialmanager.com/infd/login.asp?a=r) Please contact the publication office if you have any questions.